Author's response to reviews

Title: The pharmacokinetics of cefazolin in patients undergoing elective & semi-elective abdominal aortic aneurysm open repair surgery

Authors:

Alexandra Douglas (Alexandra_Douglas@health.qld.gov.au)
Mahdi Altukroni (mahdi_altukroni@health.qld.gov.au)
Andrew A Udy (andrew_udy@health.qld.gov.au)
Michael S Roberts (m.roberts@uq.edu.au)
Kersi Taraporewalla (Kersi_Taraporewalla@health.qld.gov.au)
Jason Jenkins (Jason_Jenkins@health.qld.gov.au)
Jeffrey Lipman (j.lipman@uq.edu.au)
Jason A Roberts (j.roberts2@uq.edu.au)

Version: 2  Date: 16 February 2011

Author's response to reviews:

16th February 2010

Dear Ms Titmus,

Thank you for your feedback and email. Your query about the Acknowledgements section which requests identification of other persons that have made:

"substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship"

All contributors on these issues are in the author list and the author contributions. The sources of funding are described in the section Statement of Funding on page 2.

To further this, we have co-located some of the above information into an Acknowledgments section as suggested. - there are no other persons requiring recognition.

Best wishes

Jason Roberts